\begin{table}%{ccccccc}%ta.2
\caption{\label{tab:lim}Photometry upper limits of SMGs.}
\begin{tabular}{lrrrrl}
\hline \hline \noalign{\smallskip}
& & $\lambda_{\rm obs}$ & Flux & Error & \\
SMG & $z$ & ($\mu$m) & ($\mu$Jy) & ($\mu$Jy) & Reference \\
\hline  
SMMJ030227.73+000653.5 & 1.408 & 70.000 & 13600.000 & 0.000
 & \citet{hainline08phd} \\ 
SMMJ030231.81+001031.3 & 1.316 & 0.365 & 0.059 & 0.000 & \citet{clements04} \\ 
SMMJ030231.81+001031.3 & 1.316 & 0.428 & 0.100 & 0.000 & \citet{chapman05} \\ 
SMMJ030231.81+001031.3 & 1.316 & 0.440 & 0.102 & 0.000 & \citet{clements04} \\ 
SMMJ030231.81+001031.3 & 1.316 & 0.656 & 0.331 & 0.000 & \citet{chapman05} \\ 
SMMJ030231.81+001031.3& 1.316& 0.767    &     0.806    &        0.000&\citet{smail04}\\
SMMJ030231.81+001031.3 &1.316& 0.767    &     0.205    &        0.000 &\citet{clements04}\\
SMMJ030231.81+001031.3 &1.316 &2.170&2.865&0.000 &\citet{smail04} \\
SMMJ030231.81+001031.3 &1.316&2.170    &     2.655    &        0.000&\citet{clements04}\\
SMMJ030231.81+001031.3 &1.316& 3.600   &      0.700   &         0.000&\citet{hainline08phd}\\
SMMJ030231.81+001031.3& 1.316 &5.800   &      5.500   &        0.000&\citet{hainline08phd}\\
SMMJ030231.81+001031.3& 1.316& 8.000    &     3.500    &        0.000& \citet{hainline08phd}\\
SMMJ030231.81+001031.3& 1.316 &24.000    &    110.700   &       0.000&\citet{hainline08phd}\\
SMMJ030231.81+001031.3& 1.316 &70.000&        14800.000 &       0.000& \citet{hainline08phd} \\
SMMJ030236.15+000817.1 &2.435 &0.365 &        0.059      &      0.000& \citet{clements04}\\
SMMJ030236.15+000817.1 &2.435 &0.428   &      0.100   &         0.000& \citet{chapman05}\\
SMMJ030236.15+000817.1 &2.435 &0.440 &        0.102     &       0.000 &\citet{clements04}\\
SMMJ030236.15+000817.1 &2.435 &0.767  &       0.205 &           0.000& \citet{clements04}\\
SMMJ030236.15+000817.1 &2.435& 0.767     &    0.806   &         0.000 &\citet{smail04}\\
SMMJ030236.15+000817.1 &2.435 &1.250   &      4.809   &         0.000& \citet{smail04}\\
SMMJ030236.15+000817.1 &2.435 &2.170  &       2.865 &           0.000& \citet{smail04}\\
SMMJ030236.15+000817.1 &2.435 &8.000        & 3.500        &    0.000& \citet{hainline08phd} \\
SMMJ030236.15+000817.1& 2.435 &24.000   &     78.200    &       0.000& \citet{hainline08phd}\\
SMMJ030236.15+000817.1 &2.435 &70.000   &     10000.000   &     0.000& \citet{hainline08phd}\\
SMMJ030238.62+001106.3 &0.276 &5.800  &       5.100  &          0.000& \citet{hainline08phd}\\
SMMJ030238.62+001106.3 &0.276 &8.000       &  3.800     &       0.000& \citet{hainline08phd}\\
SMMJ030238.62+001106.3 &0.276 &24.000       & 107.500    &      0.000&\citet{hainline08phd}\\
SMMJ030238.62+001106.3 &0.276 &70.000       & 12700.000    &    0.000 &\citet{hainline08phd}\\
SMMJ030244.82+000632.3 &0.176& 5.800     &    2.600     &       0.000 &\citet{hainline08phd}\\
SMMJ030244.82+000632.3 &0.176 &24.000   &     83.700       &    0.000 &\citet{hainline08phd}\\
SMMJ030244.82+000632.3 &0.176 &70.000        &10400.000        &0.000 &\citet{hainline08phd}\\
SMMJ105151.69+572636.0 &1.147 &70.000        &4200.000         &0.000 &\citet{hainline08phd}\\
SMMJ105151.69+572636.0 &1.147 &61182.000    & 60.000  &         0.000 &\citet{ivison02} \\
SMMJ105155.47+572312.7 &2.686 &70.000    &    3600.000  &       0.000 &\citet{hainline08phd}\\
SMMJ105158.02+571800.2 &2.239 &70.000   &     4100.000  &       0.000 &\citet{hainline08phd}\\
SMMJ105200.22+572420.2 &0.689& 24.000     &   65.100    &       0.000 &\citet{hainline08phd}\\
SMMJ105201.25+572445.7 &2.148 &0.767       &  0.243    &        0.000 &\citet{ivison02}\\
SMMJ105201.25+572445.7 &2.148 &70.000      &  3700.000 &        0.000 &\citet{hainline08phd}\\
SMMJ105207.49+571904.0 &2.689 &2.170       &  3.777 &           0.000&\citet{ivison02}\\
SMMJ105207.49+571904.0 &2.689 &2.170        & 4.541    &        0.000 &\citet{smail04}\\
SMMJ105207.49+571904.0 &2.689 &5.800    &     4.800     &       0.000&\citet{hainline08phd}\\
SMMJ105207.49+571904.0 &2.689 &8.000  &       5.600       &     0.000 &\citet{hainline08phd}\\
SMMJ105207.49+571904.0 &2.689 &70.000 &       3200.000  &       0.000&\citet{hainline08phd}\\
SMMJ105207.49+571904.0 &2.689 &1200.000&      400.000 &       800.000& \citet{kovacs06}\\
SMMJ105225.79+571906.4 &2.372 &70.000     &   3200.000 &        0.000 &\citet{hainline08phd}\\
SMMJ105227.58+572512.4 &2.142 &70.000      &  4500.000  &       0.000 &\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &0.656       &  0.034   &         0.000 &\citet{ivison05}\\
SMMJ105227.77+572218.2 &1.956 &0.767   &      0.243 &           0.000 &\citet{smail04}\\
SMMJ105227.77+572218.2 &1.956 &2.170     &    4.142     &       0.000 &\citet{smail04}\\
SMMJ105227.77+572218.2 &1.956 &3.600  &       0.500      &      0.000&\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &4.500   &      0.300     &       0.000&\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &5.800     &    4.900   &         0.000&\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &8.000    &     1.800      &      0.000 &\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &24.000  &      63.900     &      0.000&\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &70.000       & 4100.000    &     0.000&\citet{hainline08phd}\\
SMMJ105227.77+572218.2 &1.956 &350.000     &  11300.000   &  6700.000&\citet{kovacs06}\\
SMMJ105230.73+572209.5 &2.611 &70.000    &    4100.000   &      0.000 &\citet{hainline08phd}\\
SMMJ105238.19+571651.1 &1.852 &8.000   &      7.600  &          0.000&\citet{hainline08phd}\\
SMMJ105238.19+571651.1 &1.852& 70.000     &   3900.000    &     0.000 &\citet{hainline08phd}\\
SMMJ105238.30+572435.8 &3.036 &8.000     &    57.200      &     0.000& \citet{hainline08phd}\\
SMMJ105238.30+572435.8 &3.036 &70.000  &      6400.000 &        0.000&\citet{hainline08phd}\\
SMMJ105238.30+572435.8 &3.036 &61182.000  &   226.000     &     0.000 &\citet{ivison02}\\
SMMJ123549.44+621536.8 &2.203 &4.500   &      0.200    &        0.000&\citet{hainline08phd}\\
SMMJ123549.44+621536.8 &2.203 &8.000    &     1.500     &       0.000&\citet{hainline08phd}\\
SMMJ123549.44+621536.8& 2.203 &70.000  &      4100.000      &   0.000&\citet{hainline08phd}\\
SMMJ123553.26+621337.7 &2.098 &1.895    &     --0.244       &    1.228& \citet{capak04}\\
SMMJ123553.26+621337.7 &2.098 &24.000    &    70.900   &        0.000 &\citet{hainline08phd}\\
SMMJ123553.26+621337.7 &2.098 &70.000  &      2100.000     &    0.000&\citet{hainline08phd}\\
SMMJ123555.14+620901.7 &1.875 &70.000  &      2000.000  &       0.000&\citet{hainl??ine08phd}\\
SMMJ123600.10+620253.5 &2.710 &70.000   &     5500.000    &     0.000& \citet{hainline08phd}\\
SMMJ123600.15+621047.2 &1.994& 0.443   &      0.043   &         0.021&\citet{capak04}\\
SMMJ123600.15+621047.2& 1.994 &0.547  &       0.030    &        0.016&\citet{capak04}\\
SMMJ123600.15+621047.2& 1.994 &0.907 &        0.030   &         0.046& \citet{capak04}\\
SMMJ123600.15+621047.2& 1.994 &1.895  &       --2.029         &  0.928&\citet{capak04}\\
SMMJ123600.15+621047.2 &1.994 &70.000  &      3800.000  &       0.000 &\citet{hainline08phd}\\
SMMJ123606.72+621550.7 &2.416& 0.767    &     0.128   &         0.000&\citet{smail04}\\
SMMJ123606.72+621550.7 &2.416 &1.250  &       1.593    &        0.000 &\citet{smail04}\\
SMMJ123606.72+621550.7 &2.416 &1.895    &     7.125    &        3.028&\citet{capak04}\\
SMMJ123606.72+621550.7 &2.416 &70.000   &     3200.000   &      0.000&\citet{hainline08phd}\\
SMMJ123606.85+621021.4 &2.509 &24.000        &113.500         & 0.000 &\citet{hainline08phd}\\
SMMJ123606.85+621021.4 &2.509 &70.000       & 3900.000     &    0.000 &\citet{hainline08phd}\\
SMMJ123616.15+621513.7 &2.578 &0.365  &       0.028   &  0.023& \citet{capak04}\\
SMMJ123616.15+621513.7 &2.578 &1.895    &     1.741    &        0.995 &\citet{capak04}\\
SMMJ123616.15+621513.7 &2.578 &70.000  &      2000.000   &      0.000&\citet{hainline08phd}\\
SMMJ123618.33+621550.5 &1.865 &1.895    &     2.811    &        0.990 &\citet{capak04}\\
SMMJ123618.33+621550.5 &1.865 &70.000     &   3600.000  &       0.000&\citet{hainline08phd}\\
SMMJ123621.27+621708.4 &1.998& 1.250&         1.593&            0.000& \citet{smail04}\\
SMMJ123621.27+621708.4 &1.988 &70.000        &2200.000        & 0.000&\citet{hainline08phd}\\
SMMJ123622.65+621629.7 &2.466& 0.775&         0.023&  0.000& \citet{pope06}\\
SMMJ123622.65+621629.7 &2.466 &1.250         &1.593            &0.000 &\citet{smail04}\\
SMMJ123622.65+621629.7 &2.466 &1.895 &        2.170  &          0.832&\citet{capak04}\\
SMMJ123622.65+621629.7 &2.466 &70.000  &      2700.000  &       0.000&\citet{hainline08phd}\\
SMMJ123629.13+621045.8 &1.013 &70.000  &      1800.000  &       0.000&\citet{hainline08phd}\\
SMMJ123632.61+620800.1 &1.993 &70.000   &     2400.000   &      0.000&\citet{hainline08phd}\\
SMMJ123634.51+621241.0 &1.219 &1.0000000e+10& 1.0000000e+13  &  0.000& ?? \\
SMMJ123635.59+621424.1 &2.005 &70.000  &      2800.000   &      0.000& \citet{hainline08phd}\\
SMMJ123636.75+621156.1 &0.557 &24.000  &      127.800   &       0.000 &\citet{hainline08phd}\\
SMMJ123636.75+621156.1 &0.557 &70.000    &    5300.000  &       0.000&\citet{hainline08phd}\\
SMMJ123651.76+621221.3 &0.298 &70.000     &   1600.000   &      0.000&\citet{hainline08phd}\\
SMMJ123707.21+621408.1 &2.484 &0.365         &0.035           & 0.023 &\citet{capak04}\\
SMMJ123707.21+621408.1 &2.484 &70.000    &    1600.000   &      0.000&\citet{hainline08phd}\\
SMMJ123707.21+621408.1 &2.484& 1300.000  &    1400.000  &       0.000&\citet{tacconi06}\\
SMMJ123711.98+621325.7 &1.992 &1.250      &   1.593      &      0.000&\citet{smail04}\\
SMMJ123711.98+621325.7 &1.992 &1.895     &    2.498    &        1.119& \citet{capak04}\\
SMMJ123711.98+621325.7 &1.992 &70.000   &     1500.000  &       0.000& \citet{hainline08phd}\\
SMMJ123712.05+621212.3 &2.914 &0.365  &       0.050  &          0.027 &\citet{capak04}\\
SMMJ123712.05+621212.3 &2.914 &0.798  &       0.131    &        0.056& \citet{capak04}\\
SMMJ123712.05+621212.3 &2.914 &0.907   &      0.128   &         0.070&\citet{capak04} \\
SMMJ123712.05+621212.3 &2.914 &1.250    &     1.593   &         0.000&\citet{smail04}\\
SMMJ123712.05+621212.3 &2.914 &1.895   &      --0.846    &       0.977&\citet{capak04}\\
SMMJ123712.05+621212.3 &2.914 &70.000  &      1500.000  &       0.000&\citet{hainline08phd}\\
SMMJ123716.01+620323.3 &2.037 &70.000    &    6000.000     &    0.000&\citet{hainline08phd}\\
SMMJ123721.87+621035.3 &0.979 &70.000    &    3000.000   &      0.000&\citet{hainline08phd}\\
SMMJ131201.17+424208.1 &3.405 &1.0000000e+10 &1.0000000e+13   & 0.000 &?? \\
SMMJ131208.82+424129.1 &1.544 &1.0000000e+10 &1.0000000e+13   & 0.000 &?? \\
SMMJ131212.69+424422.5 &2.805 &0.428 &        0.058   &         0.000& \citet{chapman05}\\
SMMJ131212.69+424422.5 &2.805 &0.630      &   0.078     &       0.000& \citet{fomalont06}\\
SMMJ131212.69+424422.5 &2.805 &0.767    &     0.243 &           0.000 &\citet{smail04}\\
SMMJ131212.69+424422.5 &2.805 &0.920   &      0.140 &           0.000&\citet{fomalont06}\\
SMMJ131212.69+424422.5 &2.805 &1.250    &     2.524    &        0.000& \citet{smail04}\\
SMMJ131212.69+424422.5 &2.805 &5.800   &      8.200     &       0.000& \citet{hainline08phd}\\
SMMJ131212.69+424422.5 &2.805 &8.000  &       10.200 &          0.000 &\citet{hainline08phd}\\
SMMJ131212.69+424422.5 &2.805 &350.000    &   3700.000  &    4400.000& \citet{kovacs06}\\
SMMJ131215.27+423900.9 &2.565 &1.0000000e+10 &1.0000000e+13  &  0.000 &??\\
SMMJ131222.35+423814.1 &2.565 &1.0000000e+10 &1.0000000e+13  &  0.000 &??\\
SMMJ131225.20+424344.5 &1.038 &1.0000000e+10 &1.0000000e+13  &  0.000& ??\\
SMMJ131225.73+423941.4 &1.554 &5.800     &    4.900    &        0.000& \citet{hainline08phd}\\
SMMJ131225.73+423941.4 &1.554 &8.000   &      5.700     &       0.000&\citet{hainline08phd}\\
SMMJ131225.73+423941.4 &1.554 &350.000    &   14700.000  &   7400.000&\citet{kovacs06}\\
SMMJ131228.30+424454.8 &2.931 &2.170    &     3.445    &        0.000 &\citet{smail04}\\
SMMJ131231.07+424609.0 &2.713 &0.428    &     0.058     &       0.000 &\citet{chapman05}\\
SMMJ131231.07+424609.0 &2.713 &1.250  &       2.524   &         0.000&\citet{smail04}\\
SMMJ131231.07+424609.0 &2.713 &2.170         &3.445           & 0.000 &\citet{smail04}\\
SMMJ131231.07+424609.0 &2.713 &3.600     &    0.600   &         0.000&\citet{hainline08phd}\\
SMMJ131231.07+424609.0 &2.713 &4.500    &     1.000     &       0.000&\citet{hainline08phd}\\
SMMJ131231.07+424609.0 &2.713 &5.800  &       5.100     &       0.000& \citet{hainline08phd}\\
SMMJ131231.07+424609.0 &2.713 &8.000    &     6.300     &       0.000& \citet{hainline08phd}\\
SMMJ131232.31+423949.5 &2.320 &1.0000000e+10 &1.0000000e+13 &   0.000& ??\\
SMMJ131239.14+424155.7 &2.242 &1.250  &       0.481  &          0.000& \citet{smail04} \\
SMMJ131239.14+424155.7 &2.242& 5.800   &      7.700   &         0.000&\citet{hainline08phd}\\
SMMJ141741.81+522823.0 &1.150 &1.0000000e+10 &1.0000000e+13 &   0.000 &?? \\
SMMJ141742.04+523025.7 &0.661 &1.0000000e+10 &1.0000000e+13  &  0.000& ?? \\
SMMJ141750.50+523101.0 &2.128 &0.767    &     0.423     &       0.000& \citet{smail04}\\
SMMJ141750.50+523101.0 &2.128& 1.250    &     2.302   &         0.000 &\citet{smail04}\\
SMMJ141750.50+523101.0 &2.128 &3.600   &     0.300    &        0.000&\citet{hainline08phd}\\
SMMJ141750.50+523101.0 &2.128& 4.500       &  0.300      &      0.000&\citet{hainline08phd}\\
SMMJ141750.50+523101.0 &2.128 &5.800      &   1.300     &       0.000&\citet{hainline08phd}\\
SMMJ141750.50+523101.0 &2.128 &8.000   &      1.800    &        0.000&\citet{hainline08phd}\\
SMMJ141750.50+523101.0 &2.128 &24.000 &       88.800 &          0.000 &\citet{hainline08phd}\\
SMMJ141750.50+523101.0 &2.128 &70.000   &     6700.000   &      0.000 &\citet{hainline08phd}\\
SMMJ141800.40+512820.3 &1.913 &1.0000000e+10 &1.0000000e+13 &   0.000 &?? \\
SMMJ141802.87+523011.1 &2.127 &0.767     &    0.423     &       0.000&\citet{smail04}\\
SMMJ141802.87+523011.1 &2.127 &8.000       &  2.000     &       0.000&\citet{hainline08phd}\\
SMMJ141802.87+523011.1 &2.127 &24.000     &   93.200   &        0.000& \citet{hainline08phd}\\
SMMJ141802.87+523011.1 &2.127 &70.000  &      6900.000    &     0.000&\citet{hainline08phd}\\
SMMJ141809.00+522803.8 &2.712 &0.767  &       0.423    &        0.000& \citet{smail04}\\
SMMJ141809.00+522803.8 &2.712 &24.000 &       142.000    &      0.000& \citet{hainline08phd}\\
SMMJ141809.00+522803.8 &2.712 &70.000     &   13400.000  &      0.000&\citet{hainline08phd}\\
SMMJ141813.54+522923.4 &3.484 &1.250       &  2.302     &       0.000& \citet{smail04}\\
SMMJ141813.54+522923.4 &3.484& 3.600     &    0.200    &        0.000& \citet{hainline08phd}\\
SMMJ141813.54+522923.4 &3.484 &4.500    &     0.500    &        0.000& \citet{hainline08phd}\\
SMMJ141813.54+522923.4 &3.484 &5.800   &      2.100   &         0.000& \citet{hainline08phd}\\
SMMJ141813.54+522923.4 &3.484& 8.000   &      3.800   &         0.000& \citet{hainline08phd}\\
SMMJ141813.54+522923.4 &3.484 &24.000     &   169.000    &      0.000& \citet{hainline08phd}\\
SMMJ141813.54+522923.4 &3.484 &70.000     &   9000.001   &      0.000 &\citet{hainline08phd}\\
SMMJ163627.94+405811.2 &3.180 &5.800     &    3.800       &     0.000&\citet{hainline08phd}\\
SMMJ163627.94+405811.2 &3.180 &8.000     &    4.100  &          0.000&\citet{hainline08phd}\\
SMMJ163627.94+405811.2 &3.180& 24.000  &      123.800    &      0.000&\citet{hainline08phd}\\
SMMJ163627.94+405811.2 &3.180 &70.000    &    6700.000       &  0.000 &\citet{hainline08phd}\\
SMMJ163631.47+405546.9 &2.283 &0.550  &       0.161   &         0.000& \citet{ivison02}\\
SMMJ163631.47+405546.9 &2.283 &0.656   &      0.124  &          0.000&\citet{ivison02}\\
SMMJ163631.47+405546.9 &2.283& 0.767  &       0.243     &       0.000 &\citet{ivison02}\\
SMMJ163631.47+405546.9 &2.283 &0.767     &    0.106      &      0.000& \citet{smail04}\\
SMMJ163631.47+405546.9 &2.283 &2.170      &   4.142    &        0.000& \citet{smail04}\\
SMMJ163631.47+405546.9 &2.283 &70.000   &     9400.000     &    0.000& \citet{hainline08phd}\\
SMMJ163631.47+405546.9 &2.283 &1200.000 &     1100.000    &   700.000 &\citet{kovacs06}\\
SMMJ163639.01+405635.9 &1.495 &70.000  &      7800.000 &        0.000&\citet{hainline08phd}\\
SMMJ163650.43+405734.5 &2.378 &70.000   &     9100.000 &        0.000&\citet{hainline08phd}\\
SMMJ163658.19+410523.8 &2.454 &0.550    &     0.161      &      0.000 &\citet{ivison02}\\
SMMJ163658.19+410523.8 &2.454 &70.000   &     7900.000  &       0.000 &\citet{hainline08phd}\\
SMMJ163658.78+405728.1 &1.190 &70.000 &       6000.000   &      0.000& \citet{hainline08phd}\\
SMMJ163704.34+410530.3 &0.840 &5.800       &  5.700    &        0.000& \citet{hainline08phd}\\
SMMJ163704.34+410530.3 &0.840 &8.000    &     5.500        &    0.000&\citet{hainline08phd}\\
SMMJ163704.34+410530.3 &0.840 &24.000 &       174.700    &      0.000& \citet{hainline08phd}\\
SMMJ163704.34+410530.3 &0.840 &70.000      &  11600.000    &    0.000& \citet{hainline08phd}\\
SMMJ163704.34+410530.3 &0.840 &1200.000   &   800.000   &    1100.000& \citet{kovacs06}\\
SMMJ163706.51+405313.8 &2.374 &70.000   &    9500.001   &      0.000& \citet{hainline08phd}\\
SMMJ221724.69+001242.1 &0.510 &1.0000000e+10 &1.0000000e+13  &  0.000 &??\\
SMMJ221725.97+001238.9 &3.094 &0.428     &    0.058      &      0.000& \citet{chapman05}\\
SMMJ221725.97+001238.9 &3.094& 1.250  &       1.593   &         0.000& \citet{smail04}\\
SMMJ221725.97+001238.9 &3.094 &2.170    &     1.982    &        0.000& \citet{smail04}\\
SMMJ221725.97+001238.9 &3.094 &24.000  &      113.900   &       0.000& \citet{hainline08phd}\\
SMMJ221725.97+001238.9 &3.094 &70.000  &      6300.000  &       0.000 &\citet{hainline08phd}\\
SMMJ221733.02+000906.0 &0.926& 1.0000000e+10 &1.0000000e+13 &   0.000 &??\\
SMMJ221733.12+001120.2 &0.652 &70.000  &      10900.000 &       0.000 &\citet{hainline08phd}\\
SMMJ221733.91+001352.1& 2.555 &2.170 &        1.982  &          0.000&\citet{smail04}\\
SMMJ221733.91+001352.1 &2.555 &24.000    &    98.100  &         0.000& \citet{hainline08phd}\\
SMMJ221733.91+001352.1 &2.555 &70.000   &     6200.000  &       0.000& \citet{hainline08phd}\\
SMMJ221735.15+001537.2 &3.098 &5.800    &     1.900    &        0.000& \citet{hainline08phd}\\
SMMJ221735.15+001537.2 &3.098 &24.000    &    112.100  &        0.000& \citet{hainline08phd}\\
SMMJ221735.84+001558.9 &3.089 &24.000    &   132.800     &     0.000&\citet{hainline08phd}\\
SMMJ221737.39+001025.1 &2.614 &24.000 &       176.700    &      0.000&\citet{hainline08phd}\\
SMMJ221737.39+001025.1 &2.614 &70.000   &     12100.000  &      0.000& \citet{hainline08phd}\\
SMMJ221804.42+002154.4 &2.517 &1.250    &     2.099   &         0.000& \citet{smail04} \\
SMMJ221804.42+002154.4 &2.517 &8.000 &        6.500   &         0.000&\citet{hainline08phd}\\
SMMJ221804.42+002154.4 &2.517 &70.000  &      23800.000&        0.000& \citet{hainline08phd}\\
SMMJ221806.77+001245.7 &3.623 &1.250    &     6.340     &       0.000& \citet{smail04}\\
SMMJ221806.77+001245.7 &3.623& 70.000   &     18500.000  &      0.000& \citet{hainline08phd} \\
\hline
\end{tabular}
\tablefoot{When the error is equal to zero, the flux column denotes 3$\sigma$ upper limit. Otherwise -- formal flux at the position of an SMG.\\ 
 ``??'' in the reference column means that there are no upper limits for this SMG. This line is added in order to preserve the same number of objects in all tables.}
\tablebib{\citet{ivison02,ivison05,chapman03b,chapman05,capak04,clements04,egami04,greve04,smail04,fomalont06,kovacs06,laurent06,tacconi06,pope06,huynh07,hainline08phd}.} 
\end{table}